Author's response to reviews

Title: Is raloxifene associated with lower risk of mortality in postmenopausal women with vertebral fractures after vertebroplasty?: A Hospital-Based Analysis.

Authors:

Fu-Mei FMS Su (sophiesu1222@adm.cgmh.org.tw)
Ying-Chou YCC Chen (r820713@ms13.hinet.net)
Tien-Tsai TTC Cheng (tiantsai@ms21.hinet.net)
Wei-Che WCL Lin (u64lin@yahoo.com.tw)
Chun-Chung CCL Lui (lchung@adm.cgmh.org.tw)

Version: 5 Date: 19 May 2015

Author's response to reviews: see over
Response to editorial comments

1) Remove disclosures as additional files.

Ans: We had remove disclosures as additional files.

2) Reword the conclusions section of the abstract as this is grammatically incorrect.

Ans: We had reword the conclusions section of the abstract

Sincerely,

Ying-Chou Chen

Department of Rheumatology, Chang Gung Memorial Hospital, Kaohsiung Taiwan
Address: Dr. Ying Chou Chen, Department of Rheumatology, Chang Gung Memorial Hospital. 123, Ta Pei Road, Niao-Sung, Kaohsiung Hsien, Taiwan, R.O.C. TEL: 886-7-7317123; Fax: 886-7-7322402; E-mail: r820713@ms13.hinet.net